Author’s response to reviews

Title: Comparing oil based ointment versus standard practice for the treatment of moderate burns in Greece: a trial based cost effectiveness evaluation

Authors:

Vilelmine J Carayanni (bilma@otenet.gr)
Evangelia G Tsati (drtsati@yahoo.com)
Georgia A Spyropoulou (gspyropoulou@gmail.com)
Fotini N Antonopoulou (frog6048@gmail.com)
John D Ioannovich (vilmatsoura@otenet.gr)

Version: 10 Date: 16 November 2011

Author’s response to reviews:

Affiliations: Affiliation has been provided for superscript number 4.

Abstract: It has been corrected

References: I have made the corrections suggested (link and number’s correction). Also, I have corrected the term et al.

Figures: Starting with the previous submission I’ve made a great effort to minimize the white space around the images. I’m afraid this is the best I can do in view of the fact that the appropriate program is not available to me.

Typography: All typographical errors have been corrected and revisions have been made.